Author's response to reviews

Title: Adverse drug events in hospitalized children at Ethiopian University Hospital: A prospective observational study

Authors:

Tsfahun Chanie (tesfuc@gmail.com)
Bisrat Hailemeskel (bhailemeskel@howard.edu)
Negussu Mekonnen (nmekonnen@msh.org)
Getahun Paulos (getahun.paulos@ju.edu.et)
Alemayehu Berhane (alemayehuber@gmail.com)
Tsinuel Girma (tsinuel@yahoo.com)

Version: 2
Date: 15 January 2015

Author's response to reviews: see over
Dear Editor-in-Chief,

BMC Pediatrics

Please consider the enclosed manuscript entitled ‘Adverse drug events in hospitalized children at Ethiopian University Hospital: A prospective observational study’ in the BMC Pediatrics for publication.

The incidence and nature of adverse drug events (ADEs) in hospitalized children have been described scarcely in developing countries including Ethiopia. Hence, investigating adverse drug events in resource limiting set ups is essential from the local and global perspectives in medication safety. There is no previous publication in this regard in this segment of population locally. Moreover, this manuscript addresses issues related to severity and preventability of ADEs. So this will help to consider other measures and development of local guidelines for the best possible prevention of ADEs. Thus, decision and policy makers, researchers use the article for program and interventional effort. Many scientists especially those in Africa can get this paper in fruitful manner as it help to refer this for their future use.

The manuscript is suitable for publication in BMC pediatrics as it has fast publication in track record. The manuscript is original and it is not under review elsewhere.

Kindest regards,
Alemayehu Berhane Mekonnen
Clinical Pharmacy Unit, College of Medicine and Health Sciences
University of Gondar
PO. Box- 196
Email: almayehuber@gmail.com